Author’s response to reviews

Title: Graphical comparisons of relative disease burden across multiple risk factors

Authors:

John Ferguson (john.ferguson@nuigalway.ie)

Neil O'Leary (neil.oleary@nuigalway.ie)

Fabrizio Maturo (fabrizio.maturo@nuigalway.ie)

Salim Yusuf (yusufs@mcmaster.ca)

Martin O'Donnell (martin.odonnell@nuigalway.ie)

Version: 2 Date: 09 Aug 2019

Author’s response to reviews:

Thank you to the reviewer for all the additional comments. I took all recommendations - except the relabeling of Figure 1. I did however choose new labels 'Prevalence of exposure in controls' for the x-axis to remove possible confusion in the plots regarding what was meant by prevalence.